Author's response to reviews

Title: Are they really refusing to travel? A qualitative study of prehospital records

Authors:

Deborah Shaw (debbie.shaw@emas.nhs.uk)
Jane V Dyas (jane.dyas@nottingham.ac.uk)
Jo Middlemass (Jo.Middlemass@broxtowehucknall-pct.nhs.uk)
Anne Spaight (anne.spaight@emas.nhs.uk)
Maureen Briggs (mmbriggs@tiscali.co.uk)
Sarah Christopher (wyrdssiter7@btinternet.com)
A Niroshan Siriwardena (niro.siriwardena@lincoln.ac.uk)

Version: 5 Date: 19 September 2006

Author's response to reviews:

Dear Editor

Thank you for your e-mail confirming acceptance of the manuscript for publication. I have made the requested changes as follows:

Keywords section: This has been removed from the manuscript.

Section headings have been re-formatted to sentence case.

The Introduction has been renamed 'Background'

Reference numbers within the manuscript have been placed before full stops.

The discussions section has been re-titled 'Discussions and conclusions'.

An apostrophe has been added to the end of 'Authors' in the section title 'Authors' contributions'.

The Authors’ contribution section has been reformatted as suggested, using initials to identify authors

The acknowledgements section has been placed after the authors' contributions section.

'Research ethics' has been renamed 'Ethical approval' and placed at the end of the methods section.

References now include the names of all authors for each paper and 'et al' has been removed throughout.

'Box 1' has been re-titled 'Table 1' and the other tables re-labelled accordingly with appropriate amendments made to references within the text.

We trust these amendments meet your requirements and look forward to the publication of the papers.

Yours faithfully

Deborah Shaw
On behalf of the authors